Author's response to reviews

Title: Triple-A Syndrome with Prominent Ophthalmic Features and a Novel Mutation in the AAAS Gene: Case Report

Authors:
Dr Brian P Brooks (brooksb@mail.nih.gov)
Robert Kleta (kletar@mail.nih.gov)
Rafael C Caruso (rc21a@nih.gov)
Caroline Stuart (cstuart@mail.nih.gov)
Jonathan Ludlow (Jeepopto@aol.com)
Constantine A Stratakis (stratakcc1.nichd.nih.gov)

Version: 6 Date: 9 Jun 2004

Please see previous cover letter for a detailed description of the changes we made in response to reviewer comments.

We wish to add some follow-up for the patient and have re-submitted a slightly-modified manuscript in response to one of Dr. Traboulsi's very reasonable suggestions. Dr. Traboulsi raised the possibility of a cone dystrophy in the subject we describe. While we still can not disprove this possibility, the fact that the subject's best-corrected vision improved over a year time with spectacle correction and amblyopia treatment argues against a progressive retinal degeneration.